Title: Preoperative dexamethasone reduces postoperative pain, nausea and vomiting following mastectomy for breast cancer.

Authors:

Jorge Gómez-Hernández (jgomezhdz@hotmail.com)
Alba Lorena Orozco-Alatorre (lorekey29@hotmail.com)
Marisela Domínguez-Contreras (marcella_bibit7@yahoo.es)
Antonio Oceguera-Villanueva (ocegueraantonio@hotmail.com)
Salvador Gómez-Romo (sgomezromo@hotmail.com)
Andrea Socorro Alvarez-Villaseñor (andrea_surgery@hotmail.com)
Clotilde Fuentes-Orozco (clotildefuen@hotmail.com)
Alejandro González-Ojeda (avygail5@yahoo.com.mx)

Version: 6 Date: 21 December 2010

Author's response to reviews: see over
The BMC Series Editorial Production Team.

This is the last version of the manuscript:


Title: Preoperative dexamethasone reduces postoperative pain, nausea and vomiting following mastectomy for breast cancer.

Journal: BMC Cancer

MS ID: 1743615208388627

1. Only one minor correction was done in reference # 26

2. The remaining manuscript is correct.

Sincerely,

Alejandro Gonzalez-Ojeda, M.D.
Corresponding Author.